Author’s response to reviews

Title: Effect of community-based health management on the elderly health: a prospective randomized control trial from China

Authors:

Jianqian Chao (chaoseu@163.com)
Yimin Wang (89wangyimin@sina.com)
Hui Xu (ghyyxh@yahoo.com)
Qing Yu (Yuqing8886@sina.com)
Lili Jiang (lily.jiang.cn@163.com)
Lin Tian (tiandarling@163.com)
Wenyuan Xie (wenyuan218@126.com)
Pei Liu (liupeiseu@126.com)

Version: 3 Date: 6 December 2011

Author’s response to reviews:

Dear Editor,

Thanks for your suggestion. We have got the trial registration number of our research: ChiCTR-OCH-11001716. We have revised the abstract.

We would like to submit our manuscript entitled “Effect of community-based health management on older people’s Health: a prospective randomized control trial from China” for review to The BMC Health Services Research.

The manuscript is an original work and has not been submitted or is under consideration for publication in another journal. The study complies with current ethical consideration. We also confirm that all the listed authors have participated actively in the study and the authors do not have any possible conflicts of interest.

Although health management is a promising method for altering inefficiencies of the existing health service pattern, reducing medical cost and dealing with aging of population, its effectiveness is still necessary to confirm. Especially in China as a developing country, although researches on health management have been carried out, prospective randomized controlled trial for the elderly health management is rare. The BMC Health Services Research has published many high level articles on health service research. We sincerely hope you could help us publish our result to promote communications in the aspect of health management.

If you need any more information concerning the manuscript please write to me. I am looking forward to hearing from you.
Sincerely,

Jianqian Chao

Phone: 13813955976

Email: chaoseu@163.com